Author's response to reviews

Title: Delayed surgical treatment for a traumatic bilateral cervical facet joint dislocation using a posterior-anterior approach

Authors:

Takashi Shimada (shimadatakashi1204@yahoo.co.jp)
Seiji Ohtori (sohtori@faculty.chiba-u.jp)
Gen Inoue (ginoue@kitasato-u.ac.jp)
Junichi Nakamura (njonedr@yahoo.co.jp)
Izumi Nakada (izu-mel@umin.ac.jp)
Hiroshi Saiki (forza_hiro2002@yahoo.co.jp)
Song Ho Chang (s5chang@yahoo.co.jp)
Kouki Kawamura (sohtori@faculty.chiba-u.jp)
Kazuhisa Takahashi (19501114@faculty.chiba-u.jp)
Hiroshi Sugiyama (hirog1326@gmail.com)

Version: 2 Date: 22 July 2012

Author's response to reviews: see over
Dear Editor-in-Chief

We would like to submit a paper entitled “Delayed surgical treatment for a traumatic bilateral cervical facet joint dislocation using a posterior–anterior approach.” by Seiji Ohtori et al.” for possible publication in Journal of Medical Case Reports.

The material has not been published or submitted for publication elsewhere. The case is extremely rare in area of orthopedics. It is an original case report of interest to a particular clinical speciality of medicine (Orthopedics). We authors have read and provided approval for the content presented in the manuscript and for its submission to Journal of Medical Case Reports. Furthermore we did not receive any financial or did not have other relationships that might lead to a conflict of interest.

In consideration of Journal of Medical Case Reports taking action in reviewing and editing our submission, the authors hereby transfer, assign, or otherwise convey all copyright ownership to Journal of Medical Case Reports in the event that such work is published by Journal of Medical Case Reports. We would greatly appreciate your kind consideration.

Sincerely yours,

Seiji Ohtori, MD, PhD       Date: July. 22, 2012
Department of Orthopaedic Surgery, Graduate School of Medicine, Chiba University.
1-8-1 Inohana, Chuo-ku, Chiba 260-8677, Japan
TEL 81/43-226-2117   FAX 81/43-226-2116 E-mail address. sohtori@faculty.chiba-u.jp